Author’s response to reviews

Title: Analysis of risk factors of ST-segment elevation myocardial infarction in young patients

Authors:

Wang YY (tjwangyy3zx@163.com)
Liu YW (lyw660620@sina.com.cn)
Liu BJ (tj3zx@163.com)
Wang Y (wangytj3zx@163.com)
Hu XM (huxmtj3zx@163.com)
Li X (lixtj3zx@163.com)
Peng WJ (pengwjtj3zx@163.com)
JinFang Li (jinfangli@hotmail.com)
Li Tong (litong3zx@sina.com)

Version: 6 Date: 21 November 2014

Author’s response to reviews:

Dear Mr. Tacbobo,

Thank you very much for your letter and advice. We have revised the paper, and would like to re-submit it for your consideration. We have addressed the comments raised by the reviewers. This manuscript has been edited and proofread by Medjaden Bioscience Limited. Point-by-point responses are provided below.

(1) Acknowledgements

By way of a section “Acknowledgements”, please acknowledge anyone who contributed towards the article by making substantial contributions to conception, design, acquisition of data, or analysis and interpretation of data, or who was involved in drafting the manuscript or revising it critically for important intellectual content, but who does not meet the criteria for authorship. Please also include the source(s) of funding for each author, and for the manuscript preparation. Authors must describe the role of the funding body, if any, in design, in the collection, analysis, and interpretation of data; in the writing of the manuscript; and in the decision to submit the manuscript for publication. Please also acknowledge anyone who contributed materials essential for the study. If a language editor has made significant revision of the manuscript, we recommend that you acknowledge the editor by name, where possible.

Response: We thank the editor for the suggestion and have added the section of acknowledgements.

(2) Title Page

Title page should contain, at minimum, the names, institutions, countries and email addresses of all authors, and the full postal address of the submitting
Please note that the <author names/author emails addresses> in your manuscript file differ from those entered in the submission system. Please correct so they are consistent with each other.

Please indicate clearly which affiliations should be linked to which authors. by way of numerical assignments.

Response: We have added the information of the names, institutions, countries and email addresses of all authors.

(3) Please move your "Study Design" from page 1 to its appropriate section.

Response: We have moved the section of "Study Design".

Your consideration for this manuscript is highly appreciated.

We look forward to hearing from you soon.

Yours sincerely,

Professor Li T
Corresponding author
Cardiac Center, Third Central Hospital of Tian Jin, Tian Jin 300170, China
Tel.: +86-02284112130
Fax: +86-02284112130
E-mail: litong3zx@sina.com